Mon Mar 17 16:47:29 GMI 2003
Year: 03 Doy: 076
Cbserver: stueben
WEATHER COVMVENT: Mon Mar 17 16:53:26 GV 2003

Clear, cool, light south wi nd.
Mon Mar 17 16:56:27 GMI 2003 CH P Start Patro
Mon Mar 17 16:56: 39 GMVMI 2003 Pl CS Start Patro

MKI'V COMMENT: Mon Mar 17 17:00: 46 GMI 2003

Preset Ol centering.

COMMENT: Mon Mar 17 17:05:51 GMI 2003

Kai ee rebooted as an SOP mai nt enance procedure just around 1700 UT by
root (Bunny). Kaiee is now back up.

PSPT COMMENT: Mon Mar 17 17:07:41 GV 2003

Qoser vi ng

COMMENT: Mon Mar 17 17:08:49 GV 2003

Dec error, reset dec and spar resuned proper guiding.

**PI CS PROBLEM*: Non Mar 17 17:10:52 GV 2003

PI CS occulting was affected by declination excursions. Readjusted
occul ting.

Mon Mar 17 17:16:18 GMI 2003 MKI V Start Patro

**VKIV PROBLEMF*: Mon Mar 17 17:28:28 GMIr 2003

First two scans had data dropouts, about 120 deg. Cycling power on KCC to
attenpt to clear problem

Mon Mar 17 17:37:24 GMI 2003 MKI V Start Patro

PSPT COMMENT: Mon Mar 17 17:40: 41 GV 2003

Turned off filter heater per Randy’s instructions.

Mon Mar 17 17:41:49 GMr 2003 MKI V Start Patro

**VKIV PROBLEMF*: Mon Mar 17 17:44:26 GMr 2003

After cycling KCC power patrol would not start. Reset KCC. Patrol started,
scan nor nal

**x% EVENT COMMENT ****: Non Mar 17 18:00: 41 GVIT 2003

Smal |l linbflare at PA 100 from 1711 to 1741 UT.

MKIV COMVENT: Mbn Mar 17 18:02: 33 GVIT 2003

Centered OLl.

Mon Mar 17 18:02: 40 GMI 2003 PI CS Fl at

Mon Mar 17 18:02: 47 GMI 2003 CH P LSD

Mon Mar 17 18: 04: 38 GMI 2003 CH P End LSD

Mon Mar 17 18: 04: 48 GMI 2003 CH P Bi asLSD

Mon Mar 17 18:05:19 GMI 2003 Pl CS End FIl at
Mon Mar 17 18:05:44 GMI 2003 CH P End Bi asLSD
Mon Mar 17 18:05:59 GV 2003 CH P Bi as

Mon Mar 17 18: 06: 57 GMI 2003 CH P End Bi as
Mon Mar 17 18:17:21 GMI 2003 CH P End Patro
Mon Mar 17 18:21:39 GMI 2003 CH P Start Patro

**CH P PROBLEM *: Non Mar 17 18:21: 05 GV 2003

first



We were getting nultiple inmages on CH P display. This was probably due to
CH P centering its inage while guider was displaced during one of the

spar power-downs or guiding adjustrments. Stopped CH P and centered inage
manual |y using the mcronmeter controller. Restarted CHI P.

Mon Mar 17 18:29:23 GMIr 2003 MKI V End Patr ol

Mon Mar 17 18:29:32 GMr 2003 MKI V Start Cal

Mon Mar 17 18:48:52 GJr 2003 MKI V End Cal

Mon Mar 17 18:48:57 GMVI 2003 MKI V Start Patrol

MKI'V COMMENT: Mon Mar 17 18:57:55 GVI 2003

Centered OL1.
Mon Mar 17 19: 33: 50 GMI 2003 Pl CS End Patr ol
Mon Mar 17 19: 34:19 GMI 2003 CH P End Patr ol

**P| CS PROBLEMF*: Mon Mar 17 19:34:28 GMIr 2003

Experiencing filter wheel failure to go to clear position.

St oppi ng observing to swap driver cards to convince ourselves that it is
not a faulty card.

COMMENT: Mon Mar 17 19: 35:42 GV 2003

Reconfi guring donme shutter at the sane tine.

Mon Mar 17 19:36:12 GMr 2003 MKI V End Patr ol

Mon Mar 17 19:44:15 GMVIr 2003 MKI V Start Patrol

Mon Mar 17 19:44:19 GMIr 2003 CH P Start Patrol

Mon Mar 17 19:44:29 GMI 2003 PI CS Start Patrol

**P| CS PROBLEMF*: Mon Mar 17 19:45:32 GMr 2003

Filter wheel is still not homing in on clear station. It got far enough
around to allow the beamto pass thru the large hole, but not far

enough to enable the hone switch. Weel was not getting far enough around
previously to allow any light thru the clear station.

GONG COMMVENT: Mon Mar 17 19:48:36 GMVI 2003

Did GONG PM for the week.

**P| CS PROBLEMF*: Mon Mar 17 19:48:55 GMr 2003

W will be jarring the spar around to open up the optics box and renoving
the filter wheel. We will try replacing the 100 ohmresistor with a

50 ohm one as a band-aid fix unless inspection of the nechani smreveal s
a problem

Mon Mar 17 20:07:18 GMr 2003 Pl CS End Patr ol

**P| CS PROBLEMF*: Mon Mar 17 20:08: 02 GMr 2003

PI CS did not respond to the End conmand so it took a bunch of inmmges with
the covers open. It is stopped now Renpbving the filter wheel.

WEATHER COMMVENT: Mon Mar 17 20:44:11 GMIr 2003

I n broken orographic clouds.

WEATHER COMMVENT: Mon Mar 17 21:01:30 GMr 2003

In solid orographic clouds.

Mon Mar 17 21:01:33 GMI 2003 Pl CS Start Patrol
Mon Mar 17 21:03:11 GMI 2003 MKI V End Patr ol
Mon Mar 17 21:05:53 GMI 2003 Pl CS End Patr ol
Mon Mar 17 21:06: 09 GMI 2003 CH P End Patr ol

**P| CS PROBLEM*: Mon Mar 17 21:01:42 GMTI 2003
Reinstalled filter wheel mechanism Mtor was turning very slowy



in one direction conpared to the other direction. W believe that

the drive software was tinming out when driving between 6563 to CLEAR

The program now has the drive direction unchanged because of the
previous drive problem This is nowthe slow direction. CLEAR to 6563
does not time out because the stations are closer together in the current
direction. As a band aid fix, we replaced the 100-ohmresistor with a
50-ohmresistor to speed up the motor. W believe the directiona

sl uggi shness is a problemin the planetary gearing on the notor since the
not or body has been replaced and the old gearbox was not. W will be
recomendi ng that a new G obe notor p/n 5A2316 be purchased tp repl ace
the "Frankenstein” unit in there. W are in clouds now, but the filter
wheel is working nicely for now.

Mon Mar 17 21:18:15 GVIr 2003 MKI V Start Patro

Mon Mar 17 21:18:37 GMr 2003 CH P Start Patro

WEATHER COMMENT: Mon Mar 17 21:18: 35 GMI' 2003

In clear hole, restarting observations.

Mon Mar 17 21:18:50 GMI 2003 Pl CS Start Patro
MKI'V COVMENT: Mon Mar 17 21:21:13 GV 2003
Centered OLl.

WEATHER COMMENT: Mon Mar 17 21:51:13 GMI 2003

Br oken orographi ¢ cl ouds passing thru.

MKI'V COMMENT: Mon Mar 17 22:24:43 GV 2003

Centered OL.

**P| CS PROBLEMF*: Mon Mar 17 23:31:57 GMr 2003

It looks like there is a piece of lint on the PICS CCD that m ght have
fallen on the chip while the covers were open during the filter whee
repair. Patrol did not execute END and canera took a bunch of exposures
whil e the covers were open exposing the CCD to dust and lint.
Stopping to attenpt renoval of the particle.

**PSPT PROBLEMF*: Mon Mar 17 23:33:24 GMr 2003

PSPT crashed at 2310 UT nmaking jpg i mages. killed and restarted.

Mon Mar 17 23:35:50 GMI 2003 Pl CS End Patro

WEATHER COMMENT: Mon Mar 17 23:55:57 GMI' 2003

I n heavy orographic clouds and fog. C osing dones for now.

Mon Mar 17 23:56:30 GMI 2003 CH P End Patro

**P| CS PROBLEMF*: Mon Mar 17 23:56:43 GMI 2003

Buggers! We opened up PSPT, dusted off the surfaces inside the covers
around the camera, and applied conpressed air to CCD between

shutter cycles at |ongest exposures. Unable to determine if that

was effective (or made it worse) due to clouds. Standing by.

Mon Mar 17 23:59:34 GMr 2003 MKI V End Patro

**VKIV PROBLEMF*: Tue Mar 18 01:38:33 GMIr 2003

EE IR I I R I R R R I I R R I R R R R R R I R O I O O

Di sassenbl ed wrap-up for red heater power supply cable and renoved

one wap. There are now just two waps of cable inside the

wrapup. According to Darryl’s assessnent, this should reduce the occurrence
of cabl e binding between very fewto none. Tidied up and reassenbl ed



nk4 wrapup and heater system

*************************************************8

WEATHER COMMENT: Tue Mar

In heavy fog and light drizzle.

**Pl CS PROBLEM *:

Tue NMar

18 01:45:54 GJr 2003

18 01:46:18 GVIr 2003

putting insulation cover onto PICS to protect optics overnight.

Did not get to check if
| to check it tonorrow
18 01:57:13 GMVr 2003

Darry

Tue NMar

17_16
17_19
17_25
17_41
17_44
17_47
17_50
17_53
17_56
17_59
1802
18_05
1808
18 11
18”14
1817
1820
1823
1826
1832
1838
18_48
18 51
1854

Mkl V

. rawrk4
. rawrk4
. rawnk4
. rawnk4
. rawk4
. r awk4
. rawrk4
. rawrk4
. rawnk4
. rawnk4
. rawnk4
. r awk4
. rawrk4
. rawrk4
. rawnk4
. rawnk4
. rawnk4
. r awk4
. rawrk4
. rawrk4
. rawnk4
. rawnk4
. rawnk4
. r awk4

18_57.
19_00.
19_03.
19_06.
19_009.
19_12.
19_15.
19_18.
19 21.
19_24.
19_27.
19_30.
19_32.
19_44.
19_47.
19_53.
19_56.
19_509.
20_02.
20_05.
20_08.
20 11.
20_14.
20 17.

r awk4
r awk4
r awnk4
r awnk4
r awk4
r awk4
r awk4
r awk4
r awnk4
r awnk4
r awk4
r awk4
r awk4
r awk4
r awnk4
r awnk4
r awk4
r awk4
r awk4
r awk4
r awnk4
r awnk4
r awk4
r awk4

lint

r emroval

20_20.
20_23.
20_26.
20 29.
20 32.
20_35.
20_37.
20_40.
20_43.
20_46.
20_49.
20 52.
20 _55.
20 59.
2118,
21 21.
21 24.
21_27.
21_30.
21_32.
21_35.
21_38.
21 41.
21_44.

was successful .

r awk4
r awrk4
r awnk4
r awnk4
r awk4
r awk4
r awk4
r awrk4
r awnk4
r awnk4
r awk4
r awk4
r awk4
r awrk4
r awnk4
r awnk4
r awk4
r awk4
r awk4
r awrk4
r awnk4
r awnk4
r awk4
r awk4

21_47.
21 51.
21 55.
21 58.
22 01.
22 04.
22 07.
22_09.
227 12.
22”15,
22 18.
22 21.
22 24.
22 27.
22_30.
22 33.
22 36.
22 39.
22 42.
2245,
22748,
22 51.
22 54,
22 57.

r awk4
r awrk4
r awnk4
r awnk4
r awk4
r awk4
r awk4
r awrk4
r awnk4
r awnk4
r awk4
r awk4
r awk4
r awrk4
r awnk4
r awnk4
r awk4
r awk4
r awk4
r awrk4
r awnk4
r awnk4
r awk4
r awk4

Left note for

23 _00. rawnk4
23 02.rawnk4
23 _05.rawnk4
23_08. rawnk4
23 11. rawnk4
23 _14. rawnk4
23 17.rawnk4
23 20.rawnk4
23 23.rawnk4
23_26. rawrk4
23 29. rawnk4
23 _32. rawnk4
23 35.rawnk4
23 38.rawnk4
23_41. rawrk4
23_44. r awrk4
23 _47. rawk4
23 _50. rawnk4
23 53.rawnk4
23 56. rawnk4
c18 29.rawnk4
c18_35. rawnk4
cl1l8 41.rawnk4



